Author's response to reviews

Title: Social and economic burden of walking and mobility problems in multiple sclerosis

Authors:

James Pike (james.pike@adelphigroup.com)
Edward Jones (eddie.jones@adelphigroup.com)
Krithika Rajagopalan (krithika.rajagopalan@sunovion.com)
James Piercy (james.piercy@adelphigroup.com)
Peter Anderson (peter.anderson@adelphigroup.com)

Version: 3 Date: 6 August 2012

Author's response to reviews: see over
Dear Deesha,

Re: MS: 2077999172613700  
Social and economic burden of walking and mobility problems in multiple sclerosis: results of a cross-sectional study in Europe James S Pike, Eddie C Jones, Krithika Rajagopalan, James Piercy and Peter R Anderson BMC Neurology

Further to your email dated 31st July 2012 I am pleased to attach a revised version of the above paper which we believe addresses the editorial requests outlined (in said e-mail).

Now included after the ‘Discussion’ section are details regarding ‘Competing Interests’, ‘Author Contributions’ and ‘Acknowledgements’ which can be found on page 15 of 23 in the paper.

Thank you for you invitation to resubmit the paper to BMC Neurology and look forward to hearing from you.

With kind regards,

James